Author’s response to reviews

Title: Public awareness and perception of Ghana's restrictive policy on fatty meat, as well as preference and consumption of meat products among Ghanaian adults living in the Kumasi Metropolis

Authors:

Reginald Annan (reggie@imtf.org)

Charles Apprey (apreycharles@yahoo.com)

Kwasi Oppong (kwasioppong2000@yahoo.com)

Vanessa Petty-Agamatey (vanispa@yahoo.com)

Laudina Mensah (lauddymens@yahoo.co.uk)

Anne Marie Thow (annemarie.thow@sydney.edu.au)

Version: 2 Date: 23 Nov 2017

Author’s response to reviews:

Dear Editor

I wish to submit a revised manuscript following the comments in the email sent to us on 14th November 2017.

We have made all the revisions requested, including, providing email addresses of all authors on the front page, adding consent to publish sections, providing information on funding, adding missing author, providing a list of abbreviations and a single, final, clean version of the manuscript that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours.

Thank you for the opportunity to publish in BMC Nutrition Journal.

Sincerely,

Reginald A. Annan